Author’s response to reviews

Title: Stable clinical course in three siblings with late-onset isolated sulfite oxidase deficiency: a case series and literature review.

Authors:

Maoqiang Tian (drmaoqiang@126.com)
Yi Qu (quyi712002@163.com)
Lingyli Huang (2017151642114@stu.scu.edu.cn)
Xiaojuan Su (xiaojuansu2017@163.com)
Shiping Li (iamlsp@163.com)
Junjie Ying (yingjunjie177@163.com)
Fengyan Zhao (zhaofengyan621@163.com)
Dezhi Mu (mud@scu.edu.cn)

Version: 4 Date: 11 Dec 2019

Author’s response to reviews:

December 11, 2019

Dear editors,

Thank you very much for consideration of our manuscript “Stable clinical course in three siblings with late-onset isolated sulfite oxidase deficiency: a case series and literature review” (BPED-D-19-00217_R3). We appreciate the thoughtful comments of the editor and reviewers, as current suggestions and comments are valuable for improving our manuscript. We have revised the manuscript to reflect their criticisms.

Sincerely,

Yi Qu

Prof. Yi Qu
Department of Pediatrics,
West China Second University Hospital,
Response to reviewers

We appreciate the thoughtful comments of the reviewers, and we have revised the manuscript to reflect their criticisms.

Editor Comments:

1-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We appreciate the reviewer’s comments. We have re-edited the section of Funding per reviewer’s comments.

2-- In the Abstract, please replace the word "abnormities" to abnormalities.

Response: We appreciate the reviewer’s suggestion. We have used “abnormalities” instead of the term " abnormities " in the Abstract section.

3-- Thank you for providing an editing certificate. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We appreciate the reviewer’s comments. We have removed editing certificate in this submission. We have uploaded the manuscript as a single, final, clean version. Thus, tables, figures also were clean versions in this submission.